Author’s response to reviews

Title: Digital versus analogue chest drainage system in patients with primary spontaneous pneumothorax: a randomized controlled trial

Authors:

Kris Mooren (k.mooren@spaarnegasthuis.nl)
Dieuwertje Ruigrok (g.a.ruigrok@umcutrecht.nl)
Peter Kunst (p.w.a.kunst@olvg.nl)
Marielle Blacha (m.blacha@spaarnegasthuis.nl)
Ben Tomlow (be.tomlow@isala.nl)
Jacobine Herbrink (j.herbrink@vanweelbethesda.nl)
Eva Japenga (ejapenga@gmail.com)
Wim Boersma (w.boersma@nwz.nl)
Paul Bresser (p.bresser@olvg.nl)
Ivo van der Lee (ivanderlee@spaarnegasthuis.nl)

Version: 2 Date: 27 Apr 2020

Author’s response to reviews:

Dear editor, dear Ms. Clayton,

thank you for the opportunity to publish in your journal, and for the thorough revision of our manuscript.

Your comment was:
1. Under your Ethics approval and consent to participate heading in the Declarations section, please clarify whether informed consent obtained from participants was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.

The informed consent was written. This has been adjusted in the text. There are no further adjustments to the manuscript.

We look forward to hearing from you in due time regarding our submission, and to respond to any further questions and comments you may have.

Best regards on behalf of all authors,

Kris Mooren